Author’s response to reviews

Title: Obesity and pregnancy: a transversal study from a low-risk maternity

Authors:

Ana Carolina S Calderon (ana_carol1@hotmail.com)
Silvana M Quintana (quintana@fmrp.usp.br)
Alessandra C Marcolin (dralemar@uol.com.br)
Aderson T Berezovski (atberezov@fmrp.usp.br)
Luiz Gustavo O Brito (lgobrito@gmail.com)
Geraldo Duarte (gduarte@fmrp.usp.br)
Ricardo C Cavalli (rcavalli@fmrp.usp.br)

Version: 4
Date: 18 July 2014

Author’s response to reviews: see over
It is with pleasure that we are re-sending a manuscript (Obesity and pregnancy: a transversal study from a low-risk maternity) after review for the analysis of your editorial board, intending to publish it in your respectful journal.

We have included the editorial requests in this cover letter.

(1) Please include a list of abbreviations.
R=Done, it is located after conclusions, according to the authors’ guidelines.

(2) Acknowledgments
- By way of a section ?Acknowledgments?, please acknowledge anyone who contributed towards the article by making substantial contributions to conception, design, acquisition of data, or analysis and interpretation of data, or who was involved in drafting the manuscript or revising it critically for important intellectual content, but who does not meet the criteria for authorship. Please also include the source(s) of funding for each author, and for the manuscript preparation. Authors must describe the role of the funding body, if any, in design, in the collection, analysis, and interpretation of data; in the writing of the manuscript; and in the decision to submit the manuscript for publication. Please also acknowledge anyone who contributed materials essential for the study. If a language editor has made significant revision of the manuscript, we recommend that you acknowledge the editor by name, where possible. The role of a scientific (medical) writer must be included in the acknowledgements section, including their source(s) of funding. We suggest wording such as 'We thank Jane Doe who provided medical writing services on behalf of XYZ Pharmaceuticals Ltd.'

Authors should obtain permission to acknowledge from all those mentioned in the Acknowledgments section.
- Please note that if you don’t have someone to acknowledge, we would still need you to provide an Acknowledgments section within the manuscript and you may also mention there that you do not have someone to acknowledge to.
R= We decided to acknowledge in a generic way, to all women who participated at the study. I have seen that this type of acknowledgment was published in other articles from BMC Pregnancy and Childbirth and we thought it would be a good thing to do to them.

(3) Please ensure that the order in which your tables are cited is the same as the order in which they are provided. Every table must be cited in the text, using Arabic numerals. Please do not use ranges when listing tables. Tables must not be subdivided, or contain
tables within tables. Please note that we are unable to display vertical lines or text within tables, no display merged cells: please re-layout your table without these elements. Tables should be formatted using the Table tool in your word processor. Please ensure the table title is above the table and the legend is below the table. For more information, see the instructions for authors on the journal website.

R= We have formatted all tables into a same pattern, with no vertical lines among variables, with just two horizontal lines in the first row and no vertical lines. Merged cells were undone and we have used the table tool to fit them.

(4) Please remove the rebuttal as additional file and this should be uploaded as a cover letter.
R= Done.

(5) In accordance with BioMed Central editorial policies (http://www.biomedcentral.com/about/editorialpolicies#StandardsofReporting), could you please ensure your manuscript reporting adheres to STROBE guidelines(http://www.strobe-statement.org/) for reporting observational research. This is so your methodology can be fully evaluated and utilised. Can you please include a completed STROBE checklist as an additional file when submitting your revised manuscript.
R = Done, we have uploaded a Strobe checklist with our modifications at the end of each recommendation.

Sincerely,

Luiz Gustavo Oliveira Brito (on behalf of all authors)